



Limnology and Oceanography Letters 4, 2019, 62–70 © 2019 The Authors. Limnology and Oceanography published by Wiley Periodicals, Inc. on behalf of Association for the Sciences of Limnology and Oceanography. doi: 10.1002/i012.10105

### LETTER

# Particulate inorganic to organic carbon production as a predictor for coccolithophorid sensitivity to ongoing ocean acidification

#### N. A. Gafar <sup>(D)</sup>, \* B. D. Eyre, K. G. Schulz

Centre for Coastal Biogeochemistry, School of Environment Science and Engineering, Southern Cross University, Lismore, New South Wales, Australia

#### Scientific Significance Statement

Coccolithophores, calcifying phytoplankton, are a vital driver of the marine carbon cycle and mostly negatively influenced by ongoing ocean acidification. Yet, the reason for observed species and strain-specific sensitivities are elusive. Here, we identify the ratio of cellular calcification to photosynthesis as a predictor for individual sensitivity, allowing us to group species and strains accordingly.

#### Abstract

Ocean acidification (OA) can induce shifts in plankton community composition, with coccolithophores being mostly negatively impacted. This is likely to change particulate inorganic and organic carbon (PIC and POC, respectively) production, with impacts on the biological carbon pump. Hence, assessing and, most importantly, understanding species-specific sensitivities of coccolithophores is paramount. In a multispecies comparison, spanning more than two orders of magnitude in terms of POC and PIC production rates, among *Calcidiscus leptoporus, Coccolithus pelagicus subsp. braarudii, Emiliania huxleyi, Gephyrocapsa oceanica,* and *Scyphosphaera apsteinii*, we found that cellular PIC : POC was a good predictor for a species' OA sensitivity. This is likely related to the need for cellular pH homeostasis, which is challenged by the process of calcification producing protons internally, especially when seawater pH decreases in an OA scenario. With higher PIC : POC, species and strains being more sensitive to OA coccolithophores may shift toward less calcified varieties in the future.

Coccolithophores are an abundant component of marine phytoplankton assemblages, which can be found in almost all ocean ecosystems ranging from the subpolar regions to the equator (McIntyre and Bé 1967). Coccolithophores produce both particulate inorganic carbon (PIC) through calcification and particulate organic carbon (POC) through photosynthesis (Rost and Riebesell 2004). The amount of carbon produced by coccolithophores varies greatly between species, with, for instance, *Emiliania huxleyi* producing less than 6 pg C cell<sup>-1</sup> and large species like *Scyphosphaera apsteinii* producing more than 200 times as much (Table 1). The ratio of inorganic to organic carbon production also varies greatly between and within species with ratios as low as 0.08 in *Chrysotila carterae* and as high as 3.29 in *Calcidiscus leptoporus* subsp. *quadriperforatus* (Table 1).

Author Contribution Statement: NG and KS collected and analyzed the data. All authors wrote the manuscript.

Data Availability Statement: Data are available in the Pangaea repository at https://doi.pangaea.de/10.1594/PANGAEA.895158.

Additional Supporting Information may be found in the online version of this article.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

<sup>\*</sup>Correspondence: n.gafar.10@student.scu.edu.au

Associate editor: Jeffrey Krause

Species	Total	Carbon quota inorganic	Organic	PIC : POC	Size (µm)	Volume (µm³)	SA : Vol
Calculated sensitivity							
C. leptoporus <sup>H</sup>	30.3–212 <b>(106)</b>	15.8–146 <b>(60.9)</b>	14.5–82 <b>(44.3)</b>	0.50-2.21 (1.29)	11.3–14.7 <b>(12.6)</b>	1043	0.48
C. pelagicus <sup>1</sup>	355–1330 <b>(667)</b>	81–796 <b>(325)</b>	139–1071 <b>(347)</b>	0.11-2.57 (1.15)	10.5–19.9 <b>(16.0)</b>	2140	0.38
E. huxleyi <sup>t</sup>	5.9-45.6 <b>(18.4)</b>	2.9–21.3 <b>(7.2)</b>	2.5-34.2 (11.3)	0.24–1.38 (0.67)	4.5–5.1 <b>(4.8)</b>	58.1	1.25
G. oceanica <sup>H</sup>	15.5–151 <b>(45.9)</b>	7.5–90.8 <b>(24.5)</b>	7.3-73.5 (21.5)	0.52-2.44 (1.25)	6.1	119	0.98
S. apsteinii <sup>l</sup>	506–1269 <b>(844)</b>	170–811 <b>(381)</b>	317–679 <b>(465)</b>	0.42–1.44 (0.83)	12.0–22.4 <b>(17.2)</b>	2678	0.35
Estimated sensitivity							
C. leptoporus 1168 <sup>H</sup>	377–525 <b>(468)</b>	251–382 <b>(308)</b>	126–224 <b>(160)</b>	1.37–3.13 <b>(2.08)</b>	19.7	4003	0.30
C. leptoporus 1141 <sup>H</sup>	122–233 <b>(183)</b>	93.4–153 <b>(124)</b>	28.7-81.3 (58.5)	0.72-3.29 (2.01)	13.3	1232	0.45
C. leptoporus 1130 <sup>L</sup>	81.6–119 <b>(98.5)</b>	29.1–58.0 <b>(40.9)</b>	52.5-61.0 <b>(57.6)</b>	0.51-0.75 (0.61)	12.1	928	0.50
H. carteri 1323 <sup>H</sup>	401–562 <b>(481)</b>	278–384 <b>(331)</b>	124–178 <b>(151)</b>	2.29–2.30 (2.30)	14.7	1646	0.41
H. carteri 1334 <sup>H</sup>	450–614 <b>(532)</b>	290–379 <b>(334)</b>	161–235 <b>(198)</b>	1.59–1.82 (1.71)	15.8	2080	0.38
C. carterae <sup>L</sup>	75.7–522 <b>(219)</b>	7.9–59.7 <b>(21.5)</b>	60.9–463 <b>(197)</b>	0.08-0.27 (0.17)	3.8–11.7 <b>(8.2)</b>	284	0.74
S. pulchra <sup>L</sup>	121–174 <b>(156)</b>	20.9–30.0 <b>(24.2)</b>	98.6–145 <b>(131)</b>	0.15-0.22 (0.19)	9.5-10.3 <b>(9.9)</b>	511	0.60
U. sibogae <sup>L/I</sup>	43.9	16.7	27.2	0.62	N/A	N/A	N/A
Data compiled from Ba et al. (2014), Bach et a calculated using the mo	lch et al. (1992), Lange . (2015), Šupraha et al. :an cell diameter, and C.	r et al. (2006), Rahbari (2009), Fi (2015), Zhang et al. (2015), Gat <i>leptoporus</i> strains from Diner et a	iorini et al. (2011 <i>a,b</i> ), k ar and Schulz (2018), a al. (2015) were estimate	rug et al. (2011), Heinle nd N. A. Gafar et al. (20 cd, whereas those from L	: (2013), Daniels et al. ( 19). Note: volume and anger et al. (2006) werr	2014), Diner et al. (J surface area to volum e calculated. <sup>L/I/H</sup> den	2015), Set ie ratio are ote low/in

**Table 1.** Range and mean (bold) values of carbon production quotas consisting of total particulate carbon, PIC, and POC (pg C cell<sup>-1</sup>), PIC : POC ratios, and

Gafar et al.

63

Production of POC and PIC at the community level drives facets of the so-called biological carbon pump (Riebesell et al. 2009) and can be split into contributions to the organic carbon and carbonate counter pumps, respectively (Westbroek et al. 1993). Production of POC at the surface ocean and subsequent transport to depth increases the surface ocean uptake capacity for atmospheric  $CO_2$ , whereas the carbonate counter pump has the opposite effect (Rost and Riebesell 2004).

Carbon production and growth rates of coccolithophores are significantly affected by environmental conditions (i.e., temperature, light, and CO<sub>2</sub>; Langer et al. 2006; Bach et al. 2011; Sett et al. 2014; Zhang et al. 2015; N. A. Gafar et al. 2019). Human-induced perturbations to the global carbon cycle, namely increasing concentrations of atmospheric CO<sub>2</sub> will drive changes in ocean surface temperatures, light availability, and carbonate chemistry speciation (Rost and Riebesell 2004). The latter process is termed ocean acidification and leads to reductions in oceanic pH and carbonate ion concentration, whereas CO<sub>2</sub> and bicarbonate concentrations increase (Doney et al. 2009). Within a physiologically broad CO<sub>2</sub>/pH range, simulating OA, optimum curve responses for PIC and POC production, and growth rates have been observed in all coccolithophores so far (Langer et al. 2006; Bach et al. 2011; Sett et al. 2014; Zhang et al. 2015). Sensitivities of PIC, POC, and growth rates to OA vary as a result of differing substrate use (Kottmeier et al. 2016), with those for PIC production being the most sensitive (Bach et al. 2015; Gafar and Schulz 2018; N. A. Gafar et al. 2019). As a result, OA influences the relative production of PIC and POC in coccolithophores. Sensitivities to OA also vary between species with some such as G. oceanica being much more sensitive than others like E. huxleyi and S. apsteinii (Gafar et al. 2018b; Gafar and Schulz 2018; N. A. Gafar et al. 2019). It is not yet clear what drives these differences; however, they are likely to result in shifts in coccolithophore community composition under climate change conditions (Rost and Riebesell 2004; Gafar and Schulz 2018). Through effects on both community composition and physiological rates, OA has the potential to impact PIC and POC production on a global scale.

Given the potential effects of OA on coccolithophoriddriven biogeochemical element cycling and feedbacks to Earth's climate system outlined above, it becomes paramount to not only assess species-specific sensitivities but also to start investigating whether there are common underlying cellular characteristics. In the present study, we therefore examine how differences in cell size-related parameters (i.e., surface area and volume) and variation in the relative production of PIC and POC may influence species-specific sensitivities to high CO<sub>2</sub>.

#### Methods

#### Species comparison

Previously published data on responses of calcification, photosynthesis, and growth to simulated ocean acidification

(increasing CO<sub>2</sub> and decreasing pH) were collected for E. huxleyi (Sett et al. 2014; Gafar and Schulz 2018), G. oceanica (Sett et al. 2014; Zhang et al. 2015; Gafar et al. 2018b), S. apsteinii (N. A. Gafar et al. 2019), C. leptoporus (Langer et al. 2006), and Coccolithus pelagicus subsp. braarudii strains RCC1198 (Bach et al. 2015) and RCC1200 (Krug et al. 2011). All clones were isolated between 1992 and 2001 except C. pelagicus subsp. braarudii strain RCC1198, which was isolated in 1958. The fCO<sub>2</sub> level required to suppress rates to half of maximum rates ( $K_{2_{fCO_2}}^1$  inhib), or in other words, the sensitivity to OA was calculated from fits of the data to equations describing the modulating effects of temperature (T), light (I), and carbonate chemistry (substrate:  $S = CO_2$  and  $HCO_3^-$  and  $H = [H^+]$ ) on metabolic rates (MR) developed elsewhere (Gafar et al. 2018b), with Eq. 1 for species where carbonate chemistry conditions varied at constant light and temperature (C. leptoporus and C. pelagicus), Eq. 2 for species where carbonate chemistry and light intensities varied at a constant temperature (S. apsteinii), and Eq. 3 for species where carbonate chemistry, temperature, and light intensities varied (G. oceanica and E. huxleyi).  $K_{2_{fCO_2}}^1$  inhib values were calculated for each species and individual light and temperature condition. Using these half-inhibition fCO<sub>2</sub> levels is an effective way to normalize differences in absolute rates between species and between different temperature and light conditions.

$$MR(S,H) = \frac{k_1 S}{k_2 H + k_3 S H + k_4 + k_5 S}$$
(1)

$$MR(I, S, H) = \frac{k_1 SI}{k_2 H + k_3 SH + k_4 I + k_5 SI + k_6 SHI^2}$$
(2)

$$MR(T, I, S, H) = \frac{k_1 SIT}{k_2 HT + k_3 SHT + k_4 I + k_5 SI + SIT + k_6 SHI^2 T^2}$$
(3)

PIC : POC ratios were then calculated from modeled MR of calcification and photosynthesis at  $fCO_2$  levels corresponding to  $K_{2fCO_2}^1$  inhib for each rate for each species (*see* Gafar et al. [2018*a*] for calculated values). The influence of average cell size, cell volume, and surface area to volume ratio (calculated from Table 1) on OA sensitivity was also examined. Fit coefficients for *G. oceanica, E. huxleyi,* and *S. apsteinii* were taken from Gafar et al. (2018*b*), Gafar and Schulz (2018), and N. A. Gafar et al. (2019). Fit coefficients for *C. leptoporus* and *C. pelagicus* strains RCC1198 and RCC1200 are presented in Supporting Information Tables S1–S3, respectively.

#### Results

There was no clear relationship between the five species mean cell diameter ( $R^2 < 0.4135$ , F < 2.12, p > 0.24), cell volume ( $R^2 < 0.2755$ , F < 1.14, p > 0.36), or surface area : volume



**Fig. 1.** PIC : POC ratios for *E. huxleyi* (EH), *G. oceanica* (GO), *S. apsteinii* (SA), *C. leptoporus* (CL), and *C. pelagicus* (CP) vs. the *f*CO<sub>2</sub> level at which (**A**) calcification, (**B**) photosynthetic carbon fixation, and (**C**) growth rates are inhibited to half of maximum  $(K_{2CO_2}^1 \text{ inhib})$ . (I) and (T) denote treatments under varying light and temperature conditions, respectively. Numbers 1–6 denote increasing light intensities for each species. The dotted line denotes the 95% prediction bounds for new observations. Note, under some conditions, rates were so insensitive to OA that suppression to half of maximum rates did not occur within the tested limits of our fit. In these cases, no values were calculated for PIC : POC.

ratio ( $R^2 < 0.6107$ , F < 4.71, p > 0.12; Table 1) and their mean OA sensitivity. However, there was an overall linear relationship of OA sensitivity and species-specific PIC : POC ratio at the

*f*CO<sub>2</sub> of half-inhibition of calcification rates ( $R^2 = 0.7727$ , F = 71.41, p < 0.05, slope = -2380), photosynthetic rates ( $R^2 = 0.6884$ , F = 35.35, p < 0.05, slope = -4239), and growth rates ( $R^2 = 0.5793$ , F = 26.16, p < 0.05, slope = -5425), with higher ratios of PIC to POC resulting in inhibition already at lower *f*CO<sub>2</sub> (Fig. 1).

Modulation of PIC : POC ratios by light and temperature generally influenced  $K_{2fCO_2}^1$  inhib as follows. In *E. huxleyi*, an increase in light availability decreased PIC : POC ratios and hence the sensitivity to OA (higher  $K_{2fCO_2}^1$  inhib), whereas an increase in temperature had the opposite effect (Fig. 1). In *G. oceanica*, an increase in both light intensities and temperature increased both PIC : POC ratios and sensitivity to OA (lower  $K_{2fCO_2}^1$  inhib; Fig. 1). Finally, in *S. apsteinii*, there was no clear trend of light availability on PIC : POC ratios and hence sensitivity to OA (Fig. 1).

#### Discussion

#### The influence of cellular metrics on OA sensitivity

Changes in cell size-related geometry and variation in the relative production of PIC and POC both could influence a species sensitivity to OA. CaCO<sub>3</sub> formation, via calcification, is thought to result in generation of H<sup>+</sup> in the coccolith producing vesicle (Paasche 2001; Brownlee and Taylor 2004) and has been observed to result in a decrease in intracellular pH in E. huxleyi (Suffrian et al. 2011; Fig. 2). As such, more heavily calcifying species produce more H<sup>+</sup>. As coccolithophores need to maintain intracellular pH homeostasis, this excess H<sup>+</sup> must be dealt with. One option is through removal across the plasma membrane (Suffrian et al. 2011; Taylor et al. 2011). Another option is neutralization of excess  $H^+$ , for example, by reacting with  $HCO_3^-$  in the cytosol to form  $CO_2$  for photosynthesis (Fig. 2) or by reacting with hydroxyl ions (OH<sup>-</sup>) exported from the chloroplasts after internal conversion of HCO3 into CO2 (Anning et al. 1996; Buitenhuis et al. 1999; Paasche 2001). Removing/ neutralizing excess H<sup>+</sup> is likely to be more difficult for more heavily calcified species, as they must remove larger amounts of H<sup>+</sup> relative to less calcified species (Bach et al. 2015). Removing/neutralizing excess H<sup>+</sup> is also likely to be more difficult for larger species as they have a much lower surface area, relative to volume, to transport H<sup>+</sup> across. In addition to this, the removal of H<sup>+</sup> will become more difficult under high CO<sub>2</sub>/low pH conditions because of a reduced electrochemical gradient across the plasma membrane, so it would be expected that species with high PIC : POC ratios may be more quickly unable to maintain optimal intracellular pH as seawater pH levels decrease. As a result, large, high PIC : POC species would be expected to be more strongly inhibited by H<sup>+</sup> at lower CO<sub>2</sub> levels than lighter and smaller PIC : POC species.

It should be noted that the influence of PIC : POC on the OA sensitivities of growth rates is not as clear as those for calcification and photosynthetic carbon fixation rates. Interestingly,



**Fig. 2.** Schematic drawing depicting various modes of inorganic carbon uptake ( $CO_2$  and  $HCO_3^-$ ) via the plasma membrane, their mechanisms of use by photosynthetic carbon fixation in the chloroplast (Chl) and by calcification in the coccolith production vesicle (Cv) and the related H<sup>+</sup> budget. Note neutralization of excess H<sup>+</sup>, produced via calcification, can occur when photosynthesis is fuelled by  $HCO_3^-$ . While N denotes the nucleus, G depicts the Golgi vesicle. The variety of coccolith morphology in various species (e, *E. huxleyi*; u, *Umbillicosphaera* sp.; g. *Gephyrocapsa* sp.; c, *Calcidiscus* sp.; sm, *S. apsteinii* murolith; sl, *S. apsteinii* lopadolith) is illustrated by silhouettes.

it is only the larger species, of intermediate sensitivity, which do not follow the PIC : POC pattern observed for the other rates. It is possible that as cell sizes get larger that cell size begins to influence sensitivities to OA more than differences in PIC : POC (which influences small species more strongly). The reason that a similar pattern is not observed for all rates is likely because of the higher sensitivity of photosynthetic carbon fixation and calcification to OA. The higher OA sensitivity likely strengthens the influence of PIC : POC on OA sensitivity to the point that size-related effects are obscured. This would also explain why the relationship of PIC : POC and OA sensitivity becomes clearer/stronger as we move from growth (least sensitive) to photosynthetic carbon fixation (intermediate sensitivity) and finally to calcification (most sensitive). It is important to keep in mind, however, that already relatively small decreases in growth rate of much less than a 50% reduction as shown in Fig. 1C have been found and calculated to highly impact bloom formation potential and hence community production, with a ~ 50% decline in cellular standing stocks and a 1.8 times decrease in community production of CaCO<sub>3</sub> with a growth rate decrease of 10% (see Gafar et al. [2018b] and references therein).

When all five coccolithophore species are combined, a significant relationship is found between PIC : POC ratio and  $CO_2$  sensitivity, with higher PIC : POC ratios resulting in a greater sensitivity to OA in all rates (Fig. 1). This indicates that PIC : POC

ratios may be a driving factor behind species-specific sensitivities to OA, especially in terms of PIC and POC production. As  $K^{1}_{2fCO_{2}}$  inhib for all rates are generally lower (indicating higher sensitivity) in higher PIC : POC species, it would be expected that all rates will be more heavily suppressed vs. lower PIC : POC species at the same CO<sub>2</sub> level. Relative growth rates, and the photosynthetic rates which support them, are an important measure of competitive ability between species (Gafar and Schulz 2018). It has been estimated that the relatively higher decrease in growth rates of species which are more sensitive to high CO<sub>2</sub> will result in a drop in relative abundance/dominance under rising CO2 levels (i.e., G. oceanica vs. E. huxleyi in Gafar and Schulz [2018]). It is this reduction in relative competitive ability (driven by differences in PIC : POC ratios) that may shift the balance toward greater relative abundances of lower PIC : POC species under future ocean conditions.

CaCO<sub>3</sub> production in the surface ocean shifts carbonate chemistry speciation toward CO<sub>2</sub>; hence, a reduction in overall PIC production increases the storage capacity for atmospheric CO<sub>2</sub> (Zeebe and Wolf-Gladrow 2001). At the same time, it appears that PIC, through ballasting of organic carbon aggregates, is responsible for an effective mode of POC export to depth (Klaas and Archer 2002), although there are most likely other processes at work and their individual contribution, especially at the local scale are difficult to reconcile

(see Bach et al. [2016] and references therein). A reduction in PIC-driven POC export would then weaken the biological pump and decrease the ocean's storage capacity for atmospheric CO<sub>2</sub> (Barker et al. 2003). When both effects are considered together, model results suggest that the effect of reduced PIC production and reduced ballasting can nullify each other (Barker et al. 2003; Ridgwell 2003; Heinze 2004; Hofmann and Schellnhuber 2009), but the actual net effect depends on a number of assumptions, i.e., remineralization length scales, PIC : POC ballasting effect, influence of OA on PIC production (based on E. huxleyi only in most models), and sediment feedbacks (CaCO<sub>3</sub> compensation when riverine input is mismatched by deep-sea burial). Hence, what the overall effect of a shift to lower PIC : POC species, due to OA, for atmospheric CO<sub>2</sub> would be is difficult to judge at the moment.

Unlike PIC : POC ratios, cell size-related geometry does not appear to affect sensitivities and responses to changing carbonate chemistry, at least for calcification and photosynthesis. For instance, G. oceanica has similar sensitivities to C. leptoporus (798-1626 vs. 1183 µatm) even though C. leptoporus has 10 times the volume (Table 1; Fig. 1A). At the same time, E. huxleyi, which is a similar size to G. oceanica, is half as sensitive to OA (1462–3058  $\mu$ atm). It may be that the differences in size between the five species are simply not large enough in comparison to the differences in PIC : POC ratio to clearly observe a size-related effect on OA sensitivity. In addition, it may be that any effect of size which is present is being masked by the differences in PIC : POC ratio between the species (as proposed for growth rates above). Comparison of species of different sizes and similar PIC : POC ratio may be required for the presence/absence of a cell size influence to be determined.

Interestingly, the finding that calcification rates are typically more sensitive to OA than photosynthetic carbon fixation and growth rates in a single species (i.e., Sett et al. 2014; Bach et al. 2015; Zhang et al. 2015; Gafar and Schulz 2018) does also apply to our multispecies analysis. This is reflected in process-specific slopes of sensitivity to PIC : POC ratios (e.g., Fig. 1), which are the least steep (most sensitive) for calcification, followed by photosynthetic carbon fixation and growth rates (-2380, -4239, and -5425, respectively).

## The influence of temperature and light on PIC : POC ratios and high $\rm CO_2/H^+$ sensitivity

Light and temperature are known to individually modulate PIC and POC production of coccolithophores (Sett et al. 2014; Zhang et al. 2015; Gafar and Schulz 2018; N. A. Gafar et al. 2019), and in most cases, the resulting changes in PIC : POC ratio were directly reflected in corresponding shifts in  $K_{2fCO_2}^1$  inhib (i.e., following the fit line in Fig. 1A,B). This was generally the case for *E. huxleyi*, where increasing light intensities lowered PIC : POC ratios and OA sensitivity, whereas for *G. oceanica* both increased. In contrast, there was no clear trend of light on PIC : POC ratios and associated OA sensitivity in *S. apsteinii*,

and it appeared that in some instances (e.g., *G. oceanica* at  $15^{\circ}$ C and *S. apsteinii* at high light), there might be effects of light and temperature on other cellular functions, resulting in an unrelated PIC : POC and sensitivity response. It is clear that, at least, the effect of changing light intensity on PIC : POC, and therefore, sensitivities to OA, may be species specific. As a result, different species may become more or less sensitive to OA under the changes in light availability and temperature predicted over the following decades (Rost and Riebesell 2004; Stocker et al. 2013). As such, differences in temperature and light conditions should be kept in mind when comparing coccolithophore responses to OA.

## Estimating OA sensitivity using current day PIC : POC ratios

Coccolithophores can be broadly split into three groups with low (*E. huxleyi*), intermediate (*S. apsteinii* and *C. pelagicus*), and high (*G. oceanica* and *C. leptoporus*) sensitivity to OA (Fig. 1). All calcifying coccolithophores studied to date show either no change or a general decrease in PIC : POC ratio with increasing  $pCO_2$  (e.g., Krug et al. 2011; Fiorini et al. 2011*b*; Sett et al. 2014; Diner et al. 2015). As such, it should be possible to tentatively place species for which little or no  $CO_2$  response data exist, into these three general sensitivity groups using what PIC : POC ratios are available.

PIC : POC ratios of below 0.3 for *Syracosphaera pulchra* (Fiorini et al. 2011*a,b*) and *C. carterae* (Heinle 2013) suggest that these two species belong to the low-sensitivity group with *E. huxleyi*. Atlantic (RCC1323) and Mediterranean (RCC1334) strains of *Helicosphaera carteri* with average PIC : POC ratios of 2.30 and 1.71, respectively (Šupraha et al. 2015), likely belong in the high-sensitivity group with *G. oceanica*. Finally, *Umbilicosphaera sibogae*, with a PIC : POC ratio of 0.62 (Balch et al. 1992), is in between the low- and intermediate-sensitivity groups.

Concerning strain-specific sensitivities to OA, it appears that the PIC : POC ratio may also be the driving factor. For instance, calcification rates in *C. leptoporus* decreased markedly with a tripling of  $pCO_2$  in two strains with relatively high average PIC : POC ratios of 2.01 and 2.08 (RCC1141 and RCC1168; Diner et al. 2015), suggesting placement in the high-sensitivity group. In contrast, calcification and photosynthetic rates in strain RCC1130, with a comparatively low average PIC : POC ratio of 0.61, were hardly affected, placing it in the low-sensitivity group with *E. huxleyi*.

Differences in OA response between different strains have also been observed for *E. huxleyi*, with differences in response also observed between morphotypes (Langer et al. 2009; Müller et al. 2015). Morphotypes of *E. huxleyi* are physiological variants, which differ in coccolith morphology and degree of calcification of individual liths (Hagino et al. 2011). As such, it might be expected that different morphotypes may have a different degree of cellular calcification (or PIC : POC ratio) and thus sensitivity to OA. Some morphotypes with heavily calcified liths (A overcalcified) do indeed have higher cellular PIC

quotas and PIC : POC ratios than strains of morphotypes with less calcified liths (B/C) (Beaufort et al. 2011; Müller et al. 2015). At the same time, there have also been observations of no real difference in cellular PIC guota and PIC : POC ratios between different morphotypes like the R overcalcified type RCC1216, delicate B type RCC1212, and two intermediate A types RCC1238 and RCC1256 (Langer et al. 2009) and the R overcalcified types CH352 and CH360 and the moderately calcified CH428 (von Dassow et al. 2018). Interestingly, in the latter study, the similarity in cellular PIC : POC between strains/morphotypes was associated with a similar OA sensitivity. This disconnect between visual and measured degree of calcification of a strain could provide an explanation of reports of "heavily calcified" morphotypes of E. huxleyi dominating in relatively low pH waters in winter in the Bay of Biscay (Smith et al. 2012), the Benguela coastal upwelling (Henderiks et al. 2012), and the coastal zone of central Chile (Beaufort et al. 2011; von Dassow et al. 2018).

Finally, it is interesting to note that the two bloom-forming species *E. huxleyi* and *G. oceanica* belong to opposite ends of the  $CO_2$  sensitivity scale. Despite this, both species are bloom formers and the most commonly observed species in modern coccolithophore communities (McIntyre and Bé 1967; Roth and Coulbourn 1982). It is likely that current day  $CO_2$  levels are not yet high enough to impact the blooming/success of either species. As a result, OA sensitivity does not currently dictate their relative success when compared to other environmental factors like temperature (Gafar and Schulz 2018).

#### Conclusion

Combined data from *S. apsteinii, E. huxleyi, G. oceanica, C. pelagicus,* and *C. leptoporus* indicate that the PIC : POC ratio is a major driver behind species and potentially also strain-specific sensitivities of calcification, photosynthesis, and growth to OA (high  $CO_2/H^+$ ), with higher PIC : POC ratios resulting in greater sensitivity. Meanwhile, cell size, volume, and surface area to volume ratio have no clear influence on OA sensitivities. Hence, it may be expected that, under future ocean conditions, coccolithophore communities may shift toward greater abundances of lower sensitivity, low PIC : POC species and strains. This has the potential to impact the marine carbon cycle on the regional scale.

#### References

- Anning, T., N. Nimer, M. J. Merrett, and C. Brownlee. 1996.
  Costs and benefits of calcification in coccolithophorids.
  J. Mar. Syst. 9: 45–56. doi:10.1016/0924-7963(96)00015-2
- Balch, W. M., P. M. Holligan, and K. A. Kilpatrick. 1992. Calcification, photosynthesis and growth of the bloom-forming coccolithophore, *Emiliania huxleyi*. Cont. Shelf Res. **12**: 1353–1374. doi:10.1016/0278-4343(92)90059-S
- Bach, L. T., U. Riebesell, and K. G. Schulz. 2011. Distinguishing between the effects of ocean acidification and ocean

carbonation in the coccolithophore *Emiliania huxleyi*. Limnol. Oceanogr. **56**: 2040–2050. doi:10.4319/lo.2011.56.6. 2040

- Bach, L. T., U. Riebesell, M. A. Gutowska, L. Federwisch, and K. G. Schulz. 2015. A unifying concept of coccolithophore sensitivity to changing carbonate chemistry embedded in an ecological framework. Prog. Oceanogr. 135: 125–138. doi:10.1016/j.pocean.2015.04.012
- Bach, L. T., T. Boxhammer, A. Larsen, N. Hildebrandt, K. G. Schulz, and U. Riebesell. 2016. Influence of plankton community structure on the sinking velocity of marine aggregates. Global Biogeochem. Cycles **30**: 1145–1165. doi: 10.1002/2016GB005372
- Barker, S., J. A. Higgins, and H. Elderfield. 2003. The future of the carbon cycle: Review, calcification response, ballast and feedback on atmospheric CO<sub>2</sub>. Philos. Trans. R. Soc. Lond. A: Math. Phys. Eng. Sci. **361**: 1977–1999. doi:10.1098/rsta. 2003.1238
- Beaufort, L., and others. 2011. Sensitivity of coccolithophores to carbonate chemistry and ocean acidification. Nature 476: 80–83. doi:10.1038/nature10295
- Brownlee, C., and A. Taylor. 2004. Calcification in coccolithophores: A cellular perspective, p. 31–49. *In* H. R. Thierstein and J. R. Young [eds.], *Coccolithophores: From molecular processes to global impact*. Chapter 2. Springer-Verlag.
- Buitenhuis, E. T., H. J. W. de Baar, and M. J. W. Veldhuis. 1999.
  Photosynthesis and calcification by *Emiliania huxleyi* (Prymnesiophyceae) as a function of inorganic carbon species. J. Phycol. **35**: 949–959. doi:10.1046/j.1529-8817.1999. 3550949.x
- Daniels, C., R. Sheward, and A. Poulton. 2014. Biogeochemical implications of comparative growth rates of *Emiliania huxleyi* and *Coccolithus* species. Biogeosciences **11**: 6915–6925. doi:10.5194/bg-11-6915-2014
- Diner, R. E., I. Benner, U. Passow, T. Komada, E. J. Carpenter, and J. H. Stillman. 2015. Negative effects of ocean acidification on calcification vary within the coccolithophore genus *Calcidiscus*. Mar. Biol. **162**: 1287–1305. doi:10.1007/s0022 7-015-2669-x
- Doney, S. C., V. J. Fabry, R. A. Feely, and J. A. Kleypas. 2009. Ocean acidification: The other  $CO_2$  problem. Ann. Rev. Mar. Sci. **1**: 169–192.
- Fiorini, S., J. J. Middelburg, and J.-P. Gattuso. 2011a. Effects of elevated CO<sub>2</sub> partial pressure and temperature on the coccolithophore *Syracosphaera pulchra*. Aquat. Microb. Ecol. **64**: 221–232.
- Fiorini, S., J. J. Middelburg, and J.-P. Gattuso. 2011*b*. Testing the effects of elevated  $pCO_2$  on coccolithophores (Prymnesiophyceae): Comparison between haploid and diploid life stages. J. Phycol. **47**: 1281–1291.
- Gafar, N. A., and K. G. Schulz. 2018. A three-dimensional niche comparison of *Emiliania huxleyi* and *Gephyrocapsa oceanica*: Reconciling observations with projections. Biogeosciences 15: 3541–3560. doi:10.5194/bg-15-3541-2018

- Gafar, N. A., B. D. Eyre, and K. G. Schulz. 2018a, Calculated *f*CO<sub>2</sub> levels at which rates are inhibited to half of maximum and their corresponding PIC:POC ratios. PANGAEA.
- Gafar, N. A., K. G. Schulz, and B. D. Eyre. 2018b. A conceptual model for projecting coccolithophorid growth, calcification and photosynthetic carbon fixation rates in response to global ocean change. Front. Mar. Sci. **4**: 1–18.
- Gafar, N. A., B. D. Eyre, and K. G. Schulz. 2019. A comparison of species specific sensitivities to changing light and carbonate chemistry in calcifying marine phytoplankton. Sci. Rep. 9: 2486. doi:10.1038/s41598-019-38661-0
- Hagino, K., and others. 2011. New evidence for morphological and genetic variation in the cosmopolitan coccolithophore *Emiliania huxleyi* (Prymnesiophyceae) from the cox1b-atp4 genes. J. Phycol. **47**: 1164–1176. doi:10.1111/j.1529-8817. 2011.01053.x
- Heinle, M. 2013, *The effects of light, temperature and nutrients on coccolithophores and implications for biogeochemical models. PhD dissertation.* School of Environmental Sciences.
- Heinze, C. 2004. Simulating oceanic CaCO<sub>3</sub> export production in the greenhouse. Geophys. Res. Lett. **31**: L16308.
- Henderiks, J., A. Winter, M. Elbraechter, R. Feistel, A. van der Plas, G. Nausch, and R. Barlow. 2012. Environmental controls on *Emiliania huxleyi* morphotypes in the Benguela coastal upwelling system (SE Atlantic). Mar. Ecol. Prog. Ser. 448: 51–66. doi:10.3354/meps09535
- Hofmann, M., and H.-J. Schellnhuber. 2009. Oceanic acidification affects marine carbon pump and triggers extended marine oxygen holes. Proc. Natl. Acad. Sci. USA **106**: 3017–3022.
- Klaas, C., and D. E. Archer. 2002. Association of sinking organic matter with various types of mineral ballast in the deep sea: Implications for the rain ratio. Global Biogeochem. Cycles **16**: 63–1–63–14.
- Kottmeier, D. M., S. D. Rokitta, and B. Rost. 2016.  $H^+$ -driven increase in  $CO_2$  uptake and decrease in  $HCO_3^-$  uptake explain coccolithophores' acclimation responses to ocean acidification. Limnol. Oceanogr. **61**: 2045–2057.
- Krug, S., K. Schulz, and U. Riebesell. 2011. Effects of changes in carbonate chemistry speciation on *Coccolithus braarudii*: A discussion of coccolithophorid sensitivities. Biogeosciences 8: 771–777. doi:10.5194/bg-8-771-2011
- Langer, G., M. Geisen, K.-H. Baumann, J. Kläs, U. Riebesell, S. Thoms, and J. R. Young. 2006. Species-specific responses of calcifying algae to changing seawater carbonate chemistry. Geochem. Geophys. Geosyst. 7: 1–12.
- Langer, G., G. Nehrke, I. Probert, J. Ly, and P. Ziveri. 2009. Strain-specific responses of *Emiliania huxleyi* to changing seawater carbonate chemistry. Biogeosciences 6: 2637–2646. doi:10.5194/bg-6-2637-2009
- McIntyre, A., and A. W. H. Bé. 1967. Modern Coccolithophoridae of Atlantic Ocean—I. Placoliths and cyrtoliths. Deep-Sea Res. Oceanogr. Abstr. **14**: 561–597. doi:10.1016/0011-7471 (67)90065-4

- Müller, M. N., T. W. Trull, and G. M. Hallegraeff. 2015. Differing responses of three Southern Ocean *Emiliania huxleyi* ecotypes to changing seawater carbonate chemistry. Mar. Ecol. Prog. Ser. **531**: 81–90. doi:10.3354/meps11309
- Paasche, E. 2001. A review of the coccolithophorid *Emiliania huxleyi* (Prymnesiophyceae), with particular reference to growth, coccolith formation, and calcification-photosynthesis interactions. Phycologia **40**: 503–529. doi:10.2216/i0031-8884-40-6-503.1
- Rahbari, M. 2009, *Physical characteristics of Pleurochrysis carterae in relation to harvesting potential for biodiesel production. Masters thesis.* School of Chemical Engineering.
- Ridgwell, A. J. 2003. An end to the "rain ratio" reign? Geochem. Geophys. Geosyst. **4**: 1–5.
- Riebesell, U., A. Körtzinger, and A. Oschlies. 2009. Sensitivities of marine carbon fluxes to ocean change. Proc. Natl. Acad. Sci. USA **106**: 20602–20609.
- Rost, B., and U. Riebesell. 2004. Coccolithophores and the biological pump: Responses to environmental changes, p. 99–125.
   *In* H. R. Thierstein and J. R. Young [eds.], *Coccolithophores: From molecular processes to global impacts*. Chapter 5. Springer.
- Roth, P. H., and W. T. Coulbourn. 1982. Floral and solution patterns of coccoliths in surface sediments of the North Pacific. Mar. Micropaleontol. 7: 1–52. doi:10.1016/0377-8398(82)90014-7
- Sett, S., L. T. Bach, K. G. Schulz, S. Koch-Klavsen, M. Lebrato, and U. Riebesell. 2014. Temperature modulates coccolithophorid sensitivity of growth, photosynthesis and calcification to increasing seawater *p*CO<sub>2</sub>. PloS One **9**: e88308. doi: 10.1371/journal.pone.0088308
- Smith, H. E. K., and others. 2012. Predominance of heavily calcified coccolithophores at low  $CaCO_3$  saturation during winter in the Bay of Biscay. Proc. Natl. Acad. Sci. USA **109**: 8845–8849.
- Stocker, T. F., and others. [eds.]. 2013, Climate Change 2013: The physical science basis. Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge Univ. Press.
- Suffrian, K., K. G. Schulz, M. A. Gutowska, U. Riebesell, and M. Bleich. 2011. Cellular pH measurements in *Emiliania huxleyi* reveal pronounced membrane proton permeability. New Phytol. **190**: 595–608. doi:10.1111/j.1469-8137.2010.03633.x
- Šupraha, L., A. C. Gerecht, I. Probert, and J. Henderiks. 2015. Eco-physiological adaptation shapes the response of calcifying algae to nutrient limitation. Sci. Rep. 5: 16499.
- Taylor, A. R., A. Chrachri, G. Wheeler, H. Goddard, and C. Brownlee. 2011. A voltage-gated H<sup>+</sup> channel underlying pH homeostasis in calcifying coccolithophores. PLoS Biol. 9: e1001085. doi:10.1371/journal.pbio.1001085
- von Dassow, P., F. Díaz-Rosas, E. M. Bendif, J. D. Gaitán-Espitia, D. Mella-Flores, S. Rokitta, U. John, and R. Torres. 2018. Overcalcified forms of the coccolithophore *Emiliania huxleyi* in high-CO<sub>2</sub> waters are not preadapted to ocean acidification. Biogeosciences **15**: 1515–1534. doi:10.5194/bg-15-1515-2018

- Westbroek, P., and others. 1993. A model system approach to biological climate forcing. The example of *Emiliania huxleyi*. Global Planet. Change **8**: 27–46. doi:10.1016/0921-8181(93) 90061-R
- Zeebe, R. E., and D. A. Wolf-Gladrow. 2001, CO<sub>2</sub> in seawater: Equilibrium, kinetics, isotopes. Elselvier Oceanography Series, v. 65. Gulf Professional Publishing.
- Zhang, Y., L. T. Bach, K. G. Schulz, and U. Riebesell. 2015. The modulating influence of light intensity on the response of the coccolithophore *Gephyrocapsa oceanica* to ocean acidification. Limnol. Oceanogr. **60**: 2145–2157. doi:10.1002/lno.10161

#### Acknowledgments

This study was funded by the Australian Research Council FT120100384 awarded to K.G.S. and DP150102092 awarded to B.D.E. and K.G.S. We acknowledge Moritz Heinle who gave permission to use his thesis data in our analysis. We also acknowledge Ian Probert and the Roscoff Culture Collection who provided us with the original *E. huxleyi* and *S. apsteinii* cultures.

Submitted 17 August 2018 Revised 14 January 2019 Accepted 30 January 2019